Author's response to reviews

Title: The (CTG)n polymorphism in the NOTCH4 gene is not associated with schizophrenia in Japanese individuals

Authors:

Koubun Imai (zc6k-imi@asahi-net.or.jp)
Shoji Harada (sjharada@md.tsukuba.ac.jp)
Yoichi Kawanishi (qb2y-kwns@asahi-net.or.jp)
Hirokazu Tachikawa (mg8@ea.mbn.or.jp)
Takehito Okubo (ta-okubo@bc4.so-net.ne.jp)
Toshihito Suzuki (psuzukit@md.tsukuba.ac.jp)

Version: 7 Date: 24 May 2001

Dear Dr. Emma Veitch,

Six supplementary files were many versions of "Table1", and so the files were removed. I try again to submit "Table1" as a Word Table using Microsoft Word for Mac. Many thanks for your help;

Yours sincerely, Koubun Imai